

G OPEN ACCESS

Citation: Menzies SK, Thomas AO, Tianyi F-L, Abubakar SB, Nasidi A, Durfa N, et al. (2021) Livestock herding and Fulani ethnicity are a combined risk factor for development of early adverse reactions to antivenom treatment: Findings from a cross-sectional study in Nigeria. PLoS Negl Trop Dis 15(8): e0009518. https://doi.org/10.1371/ journal.pntd.0009518

Editor: Jean-Philippe Chippaux, Institut de Recherche pour le Développement, BENIN

Received: April 3, 2021

Accepted: May 29, 2021

Published: August 12, 2021

Copyright: © 2021 Menzies et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data is clinical data that contains potentially identifiable information, therefore we are required to perform some restrictions on this data - we will group some of the data (i.e. age, ethnicity) so that there are no identifiable rows. The database (one file) are deposited in the LSTM online archive repository at https://archive.lstmed.ac.uk/18503/.

RESEARCH ARTICLE

Livestock herding and Fulani ethnicity are a combined risk factor for development of early adverse reactions to antivenom treatment: Findings from a cross-sectional study in Nigeria

Stefanie K. Menzies^{1,2}, Aniekan O. Thomas^{1¤a}, Frank-Leonel Tianyi^{1,2}, Saidu B. Abubakar³, Abdulsalami Nasidi⁴, Nandul Durfa⁴, Rohit Patel^{1,2}, Anna Trelfa^{1¤b}, David G. Lalloo^{1,2}, Abdulrazaq G. Habib⁵, Robert A. Harrison^{1,2}*

 Centre for Snakebite Research and Interventions, Department of Tropical Disease Biology, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 2 Centre for Drugs and Diagnostics, Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 3 Kaltungo General Hospital, Kaltungo, Gombe State, Nigeria, 4 Special Projects Unit, Federal Ministry of Health, Abuja, Nigeria, 5 Department of Medicine, Bayero University Kano, Kano, Nigeria

¤a Current address: Department of Family Medicine, Nisa Garki Hospital, Abuja, Nigeria
¤b Current address: Greater Manchester Health Protection Team, Public Health England, Manchester, United Kingdom

* Robert.Harrison@lstmed.ac.uk

Abstract

Background

Adverse reactions to antivenom considerably complicate the clinical management of snakebite envenomed patients because it necessitates a temporary suspension of life-saving antivenom, increases costs and can compromise patient outcomes. This study sought to explore the association between cattle-herding occupation and ethnic group and the occurrence of early adverse reactions to antivenom.

Methods

This cross-sectional study was conducted between the 25th April and 11th July 2011 at the Kaltungo General Hospital in north east Nigeria. The exposure variable of cattle-herding occupation showed a strong correlation with the ethnic group variable, thus these were combined into a new variable with three categories (Fulani and herder, either Fulani or herder, and neither Fulani nor herder). The outcome variable was the occurrence of early adverse reactions, defined as any new symptoms occurring within 6 hours of antivenom administration. Odds Ratios were estimated using multivariable logistic regression models controlling for potential confounders.

Results

Among 231 envenomed snakebite victims, the overall incidence of early adverse reactions was 11.9% (95% confidence intervals: 8.0–16.9%). Patients who were Fulani and herders

Funding: The author(s) received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

had a higher incidence of early adverse reactions compared to patients who were neither Fulani nor herders (20% vs 5.7%). After adjusting for age and gender, victims who were Fulani and herders were 5.9 times more likely to have an early adverse reaction, compared to victims who were neither Fulani nor herders (95% CI: 1.88–18.59; p = 0.002).

Interpretation

To the best of our knowledge, this is the first study to provide evidence of higher odds of early adverse reactions among patients from a particular occupation and/or ethnic group. We recommend that snake envenomed patients of Fulani origin be especially closely monitored for adverse reactions, that hospitals receiving these patients be appropriately resourced to manage both envenoming and adverse reactions and that premedication with adrenaline should be considered. Our findings provide an argument for speculation on the influence of immunological or lifestyle-related differences on the occurrence of early adverse reactions to antivenom.

Author summary

Antivenom is the first-choice treatment of systemic snake envenoming that annually affects between 1.8–2.7 million victims globally. Access to antivenom is especially poor for those in greatest need because they typically reside in impoverished, rural tropical communities dependent upon health facilities with limited resources. In addition, clinical treatment of snakebite victims is further complicated by early adverse reactions (EARs) to antivenom-treatment. The causes of antivenom-associated EARs are poorly understood and under-researched. Despite antivenom producers instituting costly remedial manufacturing steps (removal of pyrogens and other impurities) to make their products safer, EARs still affect a high proportion of antivenom-treated patients. Instigated by anecdotal observations to the corresponding author from clinicians in rural Nigerian hospitals that snakebite victims of cattle-herding occupation, and especially those of Fulani ethnicity, suffer more frequent EARs than other groups, this cross-sectional study identified that risks of developing EARs to antivenom treatment include the ethnicity and pastoral lifestyle of snakebite patients. To our knowledge, this is the first study to identify that EARs to antivenom-treatment include factors associated with the victim, as well as factors related to the antivenom itself. We emphasise the need for more research on the causes of adverse reactions to antivenom so that strategies to reduce incidence can be implemented.

Introduction

Snakebite envenoming kills between 81,000–138,000 victims annually and between 1.8–2.7 million envenomed victims require treatment [1]. The first-choice treatment for envenoming is antivenom: immunoglobulins purified from the blood of horses or sheep hyper-immunised with venom. Whilst life-saving, antivenom treatment causes adverse reactions in around 20% of patients on average but, depending upon the antivenom brand (total protein content varies by manufacturer), this can be as high as 88% [2]. Whilst common, adverse reactions to antivenom are often undisclosed in publications of clinical studies, as demonstrated by Potet *et al*'s review of clinical evidence for antivenom effectiveness in sub-Saharan Africa [3], which

highlighted the value of this information to clinicians treating snakebite patients. Furthermore, a lack of consistency in definitions used to categorise severity of adverse reactions to anti-venom impedes comparison between the safety profiles of different antivenom products.

Adverse reactions to antivenom considerably complicate the clinical management of patients because it requires a temporary cessation of antivenom administration, additional medicines, resuscitation equipment and clinical expertise that are often unavailable in rural health facilities closest to most snakebite victims, and increases costs to both patients and health infrastructures. Clinicians apprehensive of adverse reactions have referred patients to other facilities, delayed administering antivenom or given suboptimal doses, which can be detrimental to patient outcomes [4].

Adverse reactions to antivenom are classified into three groups:

- 1. Early adverse reactions occur within 24 hours of antivenom administration [2], and typically manifest within the first three hours [5]. Most patients experience mild to moderate symptoms such as skin reactions (urticaria and pruritus) and gastrointestinal disturbances (vomiting, nausea, colic, diarrhoea). Severe anaphylactic reactions presenting as bronchospasm, angioedema or hypotension are less frequent but can be fatal. The true incidence of fatal reactions is unknown, under-researched, under-reported and can be mis-attributed to venom pathology [5].
- Pyrogenic reactions typically occur in the first two hours after antivenom administration and present as fever, chills, headache, myalgia, tachycardia, vasodilatation and hypotension
 These are caused by contamination of antivenom during manufacture with pyrogenic substances, such as bacterial endotoxins, protein aggregates and dimers [2,6].
- 3. Delayed or late adverse reactions ('serum sickness') occur between five to fourteen days after antivenom administration and are characterised by symptoms consistent with serum sickness observed with other antisera therapies [6]. These symptoms include gastrointestinal disturbances (nausea, diarrhoea, vomiting), urticaria, myalgia, arthralgia, pruritic rashes and lymphadenopathy [4]. Delayed serum sickness reactions to antivenom are an immuno-globulin-mediated response to exogenous immunoglobulins which activate the complement pathway [2]. Clinical evidence has shown that late adverse reactions are more likely to occur in patients previous administered serotherapy and given higher doses of antivenom [6].

Early adverse reactions are the most common category of adverse reactions and they have the greatest impact on patient care, as treatment is usually stopped following an early adverse reaction. Research evidence on the cause of these reactions is equivocal. Hypotheses include an IgE-mediated Type I hypersensitivity reaction in patients with prior exposure or sensitisation to antivenom components [7] or other animal-derived immunoglobulins (i.e. anti-tetanus and anti-rabies serotherapies), and a non-IgE-mediated activation of the complement system or heterophilic antibodies among patients without prior exposure [2,8].

There is a dearth of evidence on risk factors of early adverse reactions. Anecdotal observations from clinicians in North East Nigeria to the corresponding author suggested that cattle herders, particularly of the Fulani ethnic group, experience frequent early adverse reactions compared to the rest of the population. The Fulani are a major ethnic group in Nigeria, with an estimated population of 15.3 million, and are recognised as the dominant pastoral group in Nigeria [9], whose lifestyle, livelihood and diet are centred on cattle [10]. Fulani pastoralists generate most of their income through the sale of livestock, crops and milk [9]. This study sought to explore the association between cattle herding as an occupation, being a member of the Fulani ethnic group and the occurrence of early adverse reactions to antivenom.

Methods

Ethics statement

The study design and ethical considerations were approved by the Research Ethics Committee of the Liverpool School of Tropical Medicine on April 11, 2011 (reference number 11.31LT) and from competent authorities of Kaltungo General Hospital on April 8, 2011 (reference number GHK/ADM/58/VOL.I). Written informed consent was obtained from the participants prior to data collection. Translated versions of the consent forms were used as required, and the information sheet was read out to illiterate patients in their local language, with a thumb print used to signify consent.

Study population/setting

This cross-sectional prospective study was conducted at the Kaltungo General Hospital (KGH), in the savannah area of Gombe State, Nigeria. Most of the snakebite patients admitted to the hospital are envenomed by the West African saw-scaled viper (*Echis ocellatus*). Many patients bring the dead snake for identification and a 20-minute whole blood clotting test (20WBCT) is performed on all patients to identify patients with incoagulable blood–a distinctive symptom of envenoming by this snake in this region of Nigeria.

Patients were enrolled between 25th April and 11th July 2011. Patients were included if they were aged 18 years or older at the time of admission, they had a positive 20WBCT, they presented within 72 hours of the snakebite and they provided consent to participate in the study. Patients were excluded if they were aged 17 years and under, they had been administered antivenom at another hospital for the current snakebite, they were pregnant, or they had a severe underlying medical condition. All patients were treated with either EchiTAb-G (*E. ocellatus* monospecific, intact immunoglobulin antivenom manufactured from immunoglobulins of sheep hyper-immunised with this venom, by MicroPharm Ltd, Wales, UK) or Echi-TAb-Plus-ICP (*E. ocellatus, Bitis arietans, Naja nigricollis* tri-valent, intact immunoglobulin antivenom manufactured from immunoglobulins of horses hyper-immunised with these three venoms, by Instituto Clodomiro Picado, Costa Rica) according to local guidelines, irrespective of whether or not they were included in the study.

Following enrolment into the study, patients or their relatives were questioned by the study investigators to collect demographic data (ethnicity, age, occupation, and gender) and self-reported snakebite history (previous snakebite, antivenom received, and adverse reaction). Data was manually recorded prior to transcription onto a secure electronic database. Clinical observations and treatments (antivenom given and dose, observation of adverse reactions and symptoms) were recorded by the investigators and transcribed into the database.

Exposure variables

The exposure variables were cattle-herding occupation and Fulani or non-Fulani ethnicity. In north-eastern Nigeria, ethnicity and occupation are strongly correlated, especially for people of Fulani ethnicity. Exploratory analysis of these variables showed a strong correlation in our dataset (r = 0.5). Consequently, both variables were collapsed into binary variables, and a new variable with three categories (Fulani and livestock herder, either Fulani or herder, and neither Fulani nor herder) was created to address this collinearity.

Outcome variable

The outcome variable was the occurrence of early adverse reactions among patients treated with antivenom. Prior evidence from a randomised controlled clinical trial found no

significant difference in the risk of adverse reactions after treatment with the monovalent and trivalent EchiTAb antivenoms [11]. Consequently, no distinction between the effects of Echi-TAb-G and EchiTAb-Plus-ICP were made in our study. An early adverse reaction was defined as any new non-pyrogenic symptom (pruritus, abdominal pain, nausea, vomiting, broncho-spasm, angioedema, or hypotension) occurring within six hours following antivenom administration. The comparison group included patients who had symptoms of later reactions (between 6 hours and three days/discharge) and patients who had no reactions.

Potential confounding variables

Data was collected on demographic (age and gender) and clinical factors (initial dose, repeated doses, previous snakebite, previous antivenom use) with epidemiological evidence of an effect on the risk of early adverse reactions to antivenom.

Sample size

For a fixed sample of 231 participants, an early adverse reaction incidence of 22% (the average of previously reported incidence of EAR to EchiTAb-G (18.9%) and EchiTAb-Plus-ICP (25.8%) [11] and type I error rate of 5%, our sample had 90% power to detect an odds ratio of 3.25 between patients who were Fulani and herders and patients who were neither Fulani nor herders.

Statistical analysis

Descriptive statistics were used to summarise the demographic and clinical characteristics of the study population. Continuous variables were summarised using the median and the interquartile range, while categorical variables were summarised using frequencies and proportions. The incidence of early adverse reactions and the corresponding 95% confidence interval were calculated and presented per group of the derived exposure variable. A univariable analysis explored the association between covariables and the exposure and outcome variables. The Kruskal Wallis test was used to compare continuous and categorical variables, while chisquared and Fisher's Exact tests were used for comparing categorical variables. Potential confounders were chosen based on epidemiological and statistical (p < 0.2) bases. Multivariable logistic regression models were built using a stepwise forward regression approach. Age was included a priori into the minimally adjusted model due to strong epidemiological evidence on the association between age and early adverse reactions to antivenom. Crude and adjusted odds ratios were used to describe the association between ethnicity and occupation, and the occurrence of early adverse reactions. A likelihood ratio test was used to assess model fitness. Two-sided p-values < 0.05 were taken to indicate statistical significance. All statistical analyses were carried out using Stata v16 (StataCorp LP, College Station, TX, USA).

Results

Data from 231 victims of *E. ocellatus* (West African saw-scaled viper) envenoming who were treated with EchiTAb-G or EchiTAb-Plus-ICP antivenoms were used in this analysis. The median age was 29.0 (IQR 22.0–40.0), with a majority of victims in the 20–29 year age group. There was a male predominance (78.8%) and farming was the most common occupation (40.7%). Cattle herding was the next most common occupation and 93.6% of herders identified themselves as being of the Fulani ethnic group. Most of the victims were treated with the monovalent antivenom (78.8%) and few patients required more than one dose (5%). Table 1 presents a summary of the demographic and clinical characteristics of the study participants.

Patient characteristics	No early adverse reaction	Early adverse reaction	Total (N = 231)	P value
Age, years				0.26
Median (IQR)	30.0 (23.0-40.0)	25.0 (20.0-40.0)	29.0 (22.0-40.0)	
Age in categories				0.28
< 20	16 (8.0%)	2 (7.4%)	18 (7.8%)	
20-29	77 (38.7%)	15 (55.6%)	94 (40.7%)	
30-39	52 (25.1%)	3 (11.1%)	55 (23.8%)	
40-49	24 (12.1%)	5 (18.5%)	29 (12.6%)	
>50	28 (13.6%)	2 (7.4%)	30 (13.0%)	
Gender				0.078
Male	157 (76.4%)	25 (92.6%)	182 (78.8%)	
Female	47 (23.6%)	2 (7.4%)	49 (21.2%)	
Ethnic Group				0.008
Fulani	99 (47.2%)	21 (77.8%)	120 (51.9%)	
Hausa	11 (5.5%)	3 (11.1%)	14 (6.1%)	
Tangale	9 (4.5%)	0 (0.0%)	9 (3.9%)	
Tula	4 (2.0%)	0 (0.0%)	4 (1.7%)	
Other	80 (40.2%)	3 (11.1%)	83 (35.9%)	
Occupation				0.29
Herder	52 (24.1%)	11 (40.7%)	63 (27.3%)	
Farmer	83 (41.7%)	11 (40.7%)	94 (40.7%)	
Self employed	10 (5.0%)	1 (3.7%)	11 (4.8%)	
Student	23 (11.6%)	3 (11.1%)	26 (11.3%)	
Other	32 (15.6%)	1 (3.7%)	33 (14.3%)	
Previous snakebite				0.26
Yes	7 (3.5%)	2 (7.4%)	9 (3.9%)	
No	189 (92.5%)	22 (81.5%)	211(91.3)	
Prior antivenom use				0.14
Yes	4 (2.0%)	2 (7.4%)	6 (2.6%)	
No	191 (93.5%)	22 (81.5%)	213 (92.2%)	
Type of Antivenom				0.99
EchiTAb-G	160 (77.9%)	22 (81.5%)	182 (78.8%)	
EchiTAb-Plus-ICP	41 (20.6%)	5 (18.5%)	46 (19.9%)	
Repeated dose				
Yes	8 (3.5%)	4 (14.8%)	12 (5.2%)	0.027
No	186 (91.5%)	21 (77.8%)	207 (89.6%)	·····

Table 1. Sociodemographic and clinical characteristics of patients included in this study.

Data are presented as n(%) unless otherwise specified

https://doi.org/10.1371/journal.pntd.0009518.t001

The proportion of patients who received EchiTAb-G was 78.8%, and 19.9% received EchiTAb-Plus-ICP. The outcome of adverse reactions to antivenom was not significantly different between the two antivenoms (Table 1).

Of the 231 patients in this study, 27 had signs of an early adverse reaction following the administration of antivenom, giving an overall incidence of 11.9% (95% confidence intervals [CI]: 8.0–16.9%). The incidence was highest among Fulani herders (20%; [95% CI: 10.4–32.9]) and lowest among victims who were neither Fulani nor herder (5.7%; [95% CI: 2.1–12.1]). Mild early adverse reactions were observed in 63% of these victims, and there was no case of a severe adverse reaction.

Complete Case Analysis^ (N = 214)			Мо	del 1	Мо	del 2	Мо	del 3
	No EAR 188 (87.85)	EAR 26 (12.15)	OR	95% CI	OR	95% CI	OR	95% CI
Neither Fulani nor herder	95 (95)	5 (5)	Reference		Reference		Reference	
Either Fulani or herder	51 (83.61)	10 (16.39)	3.73*	1.20-11.48	3.56*	1.15-11.02	3.34*	1.07-10.42
Fulani and herder	42(79.25)	11 (20.75)	4.98**	1.63-15.22	4.91**	1.60-15.11	5.91**	1.88-18.59

Table 2. Multivariable analysis of the association between ethnicity/occupation and the occurrence of early adverse reaction.

 $^{:}$ Patients with missing data on ethnicity, occupation, early adverse reactions, age and gender variables were dropped; P < 0.05 *; P < 0.01**; P < 0.001***; Model 1 = Unadjusted model; Model 2 = + age; Model 3 = + gender

https://doi.org/10.1371/journal.pntd.0009518.t002

From the univariable analysis, there was no significant difference in the median age across categories of the exposure (p = 0.59) and the outcome (p = 0.125). There was a significant, male dominated difference in gender across categories of the exposure (p = 0.007) and the outcome. The remaining variables did not satisfy the statistical criteria for inclusion in the multi-variable analysis.

After adjusting for age and gender, antivenom-treated patients who were Fulani and cattleherders were 5.9 times more likely to have an early adverse reaction to antivenom, compared to patients who were neither Fulani nor herders (95% CI: 1.88–18.59; p = 0.002). Patients who were either Fulani or herders were 3.3 times more likely to have an early adverse reaction compared to patients who were neither (95% CI: 1.07–10.41; p = 0.037). Table 2 shows the multivariable analysis with a sequential adjustment for confounding.

Discussion

This study found that 11.9% of patients of *E. ocellatus* envenoming developed an early adverse reaction following antivenom treatment at the Kaltungo General Hospital. This incidence varied from a high of 20% among patients who were Fulani and herders, to a low of 5.7% among patients who were neither Fulani nor herders. Patients who were Fulani and herders were 5.9 times more likely to develop an early adverse reaction to antivenom treatment compared to patients who were neither Fulani nor herders, after accounting for age and gender. This study provides epidemiological evidence of a difference in the occurrence of early adverse reactions among snakebite victims of Fulani ethnicity in Northern Nigeria. Amongst this population, cattle-herders are at a higher risk.

To the best of our knowledge, this is the first study to provide epidemiological evidence of higher odds of early adverse reactions among patients of a particular occupation and/or ethnic background. This confirmed anecdotal reports of a higher incidence of these reactions among Fulani cattle herders. Our findings provide a basis to speculate upon the influence of immunological or lifestyle-related differences on the occurrence of early adverse reactions to antivenom treatment, and builds upon previous observations of significantly different rates of EARs to the same antivenom between urban and rural populations [12].

The Fulani group have been described as having key distinctions in levels of circulating antibodies, antibody receptors and cytokines, which may influence responses to antivenom treatment, perhaps through an atopic causation. Thus, prior studies show that Fulani have higher levels of the inflammatory cytokines IL-6, IL-8 IL-12, IFN- α and IFN- γ from an early age [13]. The Fulani group also have a higher prevalence, when compared to neighbouring ethnic groups, of an IL-4 polymorphism associated with allergy and asthma [14]. The production of IL-4 stimulates a Th2 response and class-switching of IgG1 to IgE [15], consistent with the elevated levels of total IgE observed in the Fulani[16]. Furthermore, a role has been demonstrated for the involvement of IL-4 in acute urticaria[17], whilst IL-6 can be a mediator of

pruritis [18]. It seems appropriate to speculate that the elevated levels of total IgE and inflammatory cytokines would place the Fulani at higher risk of adverse reactions to the large amounts of heterologous immunoglobulins administered during antivenom treatment. Indeed, the WHO Guidelines for the Management of Snakebites advises that patients with a history of atopic diseases are at a higher risk of severe adverse reactions to antivenom, and recommends that such patients should be appropriately pre-treated [5]. However, despite several studies reporting immunological differences between the Fulani and neighbouring groups in the context of infectious diseases, there is no data in the literature reporting the prevalence of atopy or other allergic diseases in the Fulani.

Our findings show that being both ethnic Fulani and a cattle herder was associated with higher odds of an early adverse reaction than patients who were neither Fulani nor cattle herder. This suggests that cattle exposure increases the odds of adverse reactions to antivenom within the Fulani ethnic group. The low number of non-Fulani cattle herders precluded the assessment of the cattle herding occupation as a single risk factor in non-Fulani patients. However in Europe, cattle farmers are known to have an increased risk of developing allergic reactions to cattle, with up to 20% of cattle farmers being sensitised to cattle allergens (dander, hair, meat and milk) found throughout cattle stables and farmers' homes [19]. Fulani cattle herders often keep other livestock in addition to cattle, such as sheep, goats and chickens, which may provide an additional route of sensitisation to animal allergens. To determine the influence of cattle exposure on early adverse reactions to antivenom, a future comparison between Fulani pastoralist cattle herders and other cattle-dependent ethnic groups in Africa, such as the Masai or Turkana in East Africa, would be valuable to determine the relative contribution of cattle exposure, lifestyle and diet.

Another explanation could be sensitisation to the allergenic carbohydrate galactose-alpha-1,3-galactose (α Gal) following *Amblyomma spp.* tick bites. This carbohydrate plays a central role in IgE-mediated anaphylaxis upon first exposure to the mouse-human chimeric monoclonal antibody cetuximab in α Gal-sensitised patients [20], and in IgE-mediated allergy towards mammalian meat and dairy products [21]. α Gal-sensitisation is more prevalent in populations with higher exposure to ticks, such as hunters, forest workers, and those living in rural areas [21]. The Fulani pastoralists commonly remove ticks from their cattle manually, of which Amblyomma spp. is one of the three tick genera observed in Fulani cattle [22], thus providing a potential route of α Gal-sensitisation in cattle farmers. The allergenic α Gal carbohydrate has been detected on numerous antivenoms from multiple animal species in both the whole IgG and Fab formats [23,24], and has been proposed to induce an IgE-mediated adverse reaction to antivenom [23]. The above explanations support the findings in our study and provide justification for further research on potential causes of early adverse reactions to antivenom treatment. Although one small clinical study with low power found no correlation between IgE against α Gal with increased risk of EARs to antivenom [25], we believe our findings provide a strong rationale to further investigate the contribution of α Gal sensitivity, livestock exposure and tick exposure in the development of EARs to antivenom, in a higher-powered prospective study.

The validity of our findings is supported by the fact that the study was adequately powered to detect a strong association between our exposure and outcome variables. Further, the careful creation of our exposure variable, rigorous ascertainment of the outcomes and the robustness of the statistical analysis in this study reinforces the validity of our findings. The main limitation of this study was the cross-sectional design which limits the possibility of reliably establishing a causal association. There was a possibility of selection bias given that it was a hospital-based study, it excluded patients under the age of 18 and the use of questionnaires may have introduced some recall bias. Finally, our findings are subject to residual confounding

because we did not account for all known confounders and other important factors such as time since bite and use of traditional treatment/herbal portions-which could be very high among Fulani.

The outcomes of this study suggest the need for greater clinical research on the triggers and mechanisms of antivenom-induced early adverse reactions. A rigorous clinical immunological study will define the attribution of immunogenetics and lifestyle exposures (including sensitisation to cross-reactive animal allergens and antivenom immunoglobulin allergens) upon adverse reactions to antivenom. The unusually large number of envenomed patients from varied genetic and occupational backgrounds admitted to hospitals in north eastern Nigeria provides an ideal study site for such clinical investigation and the statistical power to calculate odds ratios and detect differences in the incidence of adverse reactions to antivenom. It would be of scientific interest and clinical importance to determine whether other ethnic groups or populations dependent upon livestock agriculture, and other occupations with high levels of animal exposure, experience similarly elevated incidences of adverse reactions to antivenom.

In conclusion, we recommend that health facilities in North Eastern Nigeria take extra precautions when managing snakebite patients who are of the Fulani ethnic group and cattle herders. These patients should be appropriately treated with pre-medication and they should receive early and frequent monitoring for adverse reactions after administration of antivenom. The health facilities should be equipped with trained health care workers, essential medications and medical equipment needed to manage antivenom-induced adverse effects.

Supporting information

S1 STROBE Checklist. STROBE statement checklist for reporting of cross-sectional studies.

(DOC)

Acknowledgments

We wish to thank the patients and staff of Kaltungo General Hospital for their contributions to, and collaboration with, this study.

Author Contributions

Conceptualization: Robert A. Harrison.

Data curation: Aniekan O. Thomas, Saidu B. Abubakar, Abdulrazaq G. Habib.

- Formal analysis: Stefanie K. Menzies, Aniekan O. Thomas, Frank-Leonel Tianyi, Rohit Patel, Anna Trelfa.
- Investigation: Aniekan O. Thomas, Saidu B. Abubakar, Abdulsalami Nasidi, Nandul Durfa, Abdulrazaq G. Habib.
- **Methodology:** Aniekan O. Thomas, David G. Lalloo, Abdulrazaq G. Habib, Robert A. Harrison.
- Writing original draft: Stefanie K. Menzies, Aniekan O. Thomas, Anna Trelfa, Robert A. Harrison.
- Writing review & editing: Stefanie K. Menzies, Aniekan O. Thomas, Frank-Leonel Tianyi, Saidu B. Abubakar, Abdulsalami Nasidi, Nandul Durfa, Rohit Patel, Anna Trelfa, David G. Lalloo, Abdulrazaq G. Habib, Robert A. Harrison.

References

- Gutiérrez JM, Calvete JJ, Habib AG, Harrison RA, Williams DJ, Warrell DA. Snakebite envenoming. Nat Rev Dis Primer. 2017; 3: 17063. https://doi.org/10.1038/nrdp.2017.63
- León G, Herrera M, Segura Á, Villalta M, Vargas M, Gutiérrez JM. Pathogenic mechanisms underlying adverse reactions induced by intravenous administration of snake antivenoms. Toxicon Off J Int Soc Toxinology. 2013; 76: 63–76. https://doi.org/10.1016/j.toxicon.2013.09.010 PMID: 24055551
- Potet J, Smith J, McIver L. Reviewing evidence of the clinical effectiveness of commercially available antivenoms in sub-Saharan Africa identifies the need for a multi-centre, multi-antivenom clinical trial. PLoS Negl Trop Dis. 2019; 13: e0007551. https://doi.org/10.1371/journal.pntd.0007551 PMID: 31233536
- Offerman SR, Smith TS, Derlet RW. Does the aggressive use of polyvalent antivenin for rattlesnake bites result in serious acute side effects? West J Med. 2001; 175: 88–91. https://doi.org/10.1136/ewjm. 175.2.88 PMID: 11483547
- World Health Organization, Regional Office for South-East Asia. Guidelines for the management of snakebites. 2016.
- de Silva HA, Ryan NM, de Silva HJ. Adverse reactions to snake antivenom, and their prevention and treatment. Br J Clin Pharmacol. 2016; 81: 446–452. <u>https://doi.org/10.1111/bcp.12739</u> PMID: 26256124
- Morais V. Antivenom therapy: efficacy of premedication for the prevention of adverse reactions. J Venom Anim Toxins Trop Dis. 2018; 24: 7. https://doi.org/10.1186/s40409-018-0144-0 PMID: 29507580
- Gutiérrez JM, León G, Lomonte B, Angulo Y. Antivenoms for snakebite envenomings. Inflamm Allergy Drug Targets. 2011; 10: 369–380. https://doi.org/10.2174/187152811797200669 PMID: 21745181
- 9. Majekodunmi AO, Fajinmi A, Dongkum C, Shaw APM, Welburn SC. Pastoral livelihoods of the Fulani on the Jos Plateau of Nigeria. Pastoralism. 2014; 4: 20. https://doi.org/10.1186/s13570-014-0020-7
- Glew RH, Crossey MJ, Polanams J, Okolie HI, VanderJagt DJ. Vitamin D Status of Seminomadic Fulani Men and Women. J Natl Med Assoc. 2010; 102: 485–490. https://doi.org/10.1016/s0027-9684(15) 30556-3 PMID: 20575213
- Abubakar IS, Abubakar SB, Habib AG, Nasidi A, Durfa N, Yusuf PO, et al. Randomised Controlled Double-Blind Non-Inferiority Trial of Two Antivenoms for Saw-Scaled or Carpet Viper (Echis ocellatus) Envenoming in Nigeria. PLoS Negl Trop Dis. 2010; 4: e767. <u>https://doi.org/10.1371/journal.pntd.</u> 0000767 PMID: 20668549
- 12. Vongphoumy I, Chanthilat P, Vilayvong P, Blessmann J. Prospective, consecutive case series of 158 snakebite patients treated at Savannakhet provincial hospital, Lao People's Democratic Republic with high incidence of anaphylactic shock to horse derived F(ab')2 antivenom. Toxicon. 2016; 117: 13–21. https://doi.org/10.1016/j.toxicon.2016.03.011 PMID: 26995210
- Boström S, Giusti P, Arama C, Persson J-O, Dara V, Traore B, et al. Changes in the levels of cytokines, chemokines and malaria-specific antibodies in response to Plasmodium falciparum infection in children living in sympatry in Mali. Malar J. 2012; 11: 109. <u>https://doi.org/10.1186/1475-2875-11-109</u> PMID: 22480186
- Gupta J, Johansson E, Bernstein JA, Chakraborty R, Khurana Hershey GK, Rothenberg ME, et al. Resolving the etiology of atopic disorders by using genetic analysis of racial ancestry. J Allergy Clin Immunol. 2016; 138: 676–699. https://doi.org/10.1016/j.jaci.2016.02.045 PMID: 27297995
- 15. Junttila IS. Tuning the Cytokine Responses: An Update on Interleukin (IL)-4 and IL-13 Receptor Complexes. Front Immunol. 2018;9. https://doi.org/10.3389/fimmu.2018.00009 PMID: 29403493
- Maiga B, Dolo A, Touré O, Dara V, Tapily A, Campino S, et al. Fc gamma Receptor IIa-H131R Polymorphism and Malaria Susceptibility in Sympatric Ethnic Groups, Fulani and Dogon of Mali. Scand J Immunol. 2014; 79: 43–50. https://doi.org/10.1111/sji.12122 PMID: 24117665
- Papadopoulos J, Karpouzis A, Tentes J, Kouskoukis C. Assessment of Interleukins IL-4, IL-6, IL-8, IL-10 in Acute Urticaria. J Clin Med Res. 2014; 6: 133–137. https://doi.org/10.14740/jocmr1645w PMID: 24578755
- Song J, Xian D, Yang L, Xiong X, Lai R, Zhong J. Pruritus: Progress toward Pathogenesis and Treatment. In: BioMed Research International [Internet]. Hindawi; 11 Apr 2018 [cited 16 Aug 2020] p. e9625936. https://doi.org/10.1155/2018/9625936 PMID: 29850592
- 19. Zahradnik E, Raulf M. Animal Allergens and Their Presence in the Environment. Front Immunol. 2014;5. https://doi.org/10.3389/fimmu.2014.00005 PMID: 24478774
- Chung CH, Mirakhur B, Chan E, Le Q-T, Berlin J, Morse M, et al. Cetuximab-induced anaphylaxis and IgE specific for galactose-alpha-1,3-galactose. N Engl J Med. 2008; 358: 1109–1117. https://doi.org/10. 1056/NEJMoa074943 PMID: 18337601

- 21. Hilger C, Fischer J, Wölbing F, Biedermann T. Role and Mechanism of Galactose-Alpha-1,3-Galactose in the Elicitation of Delayed Anaphylactic Reactions to Red Meat. Curr Allergy Asthma Rep. 2019; 19: 3. https://doi.org/10.1007/s11882-019-0835-9 PMID: 30673913
- 22. Lorusso V, Picozzi K, de Bronsvoort BM, Majekodunmi A, Dongkum C, Balak G, et al. Ixodid ticks of traditionally managed cattle in central Nigeria: where Rhipicephalus (Boophilus) microplus does not dare (yet?). Parasit Vectors. 2013; 6: 171. https://doi.org/10.1186/1756-3305-6-171 PMID: 23758913
- Fischer J, Eberlein B, Hilger C, Eyer F, Eyerich S, Ollert M, et al. Alpha-gal is a possible target of IgEmediated reactivity to antivenom. Allergy. 2017; 72: 764–771. https://doi.org/10.1111/all.13073 PMID: 27775867
- 24. Straesser M, Keshavarz B, Borish L, Khokhar D, Holian A, Charlton NP, et al. α-Gal on Crotalidae-polyvalent Fab antivenom (CroFab): Investigating the relevance to immediate hypersensitivity reactions. J Allergy Clin Immunol Pract. 2021; 9: 1015–1017.e1. https://doi.org/10.1016/j.jaip.2020.10.026 PMID: 33122101
- 25. Blessmann J, Hanlodsomphou S, Santisouk B, Choumlivong K, Soukhaphouvong S, Chanthilat P, et al. Serum IgE against galactose-alpha-1,3-galactose is common in Laotian patients with snakebite envenoming but not the major trigger for early anaphylactic reactions to antivenom. Toxicon X. 2020;7. https:// doi.org/10.1016/j.toxcx.2020.100054 PMID: 32776003